Author's response to reviews

Title: Candidate high myopia loci on chromosomes 18p and 12q do not play a major role in susceptibility to common myopia

Authors:

Grace Ibay (ibayg@mail.nih.gov)
Betty Doan (bdoan@jhsph.edu)
Lauren Reider (lreider@mail.med.upenn.edu)
Debra Dana (ddana@mail.med.upenn.edu)
Melissa Schlifka (schlifka@mail.med.upenn.edu)
Heping Hu (hhu@cccb.upenn.edu)
Taura Holmes (tholmes@mail.nih.gov)
Jennifer O'Neill (joneill@jhsph.edu)
Robert Owens (stamboli@mail.med.upenn.edu)
Elise Ciner (eciner@pco.edu)
Joan Bailey-Wilson (jebw@mail.nih.gov)
Dwight Stambolian (stamboli@mail.med.upenn.edu)

Version: 5 Date: 27 July 2004

Author's response to reviews:

WE have cropped the figures, removed color from the figures and made sure changes are not highlighted.